Author's response to reviews

Title: Prevalence and sociodemographic correlates of stunting, underweight, and overweight among Palestinian school adolescents (13-15 years) in two major governorates in the West Bank

Authors:

Nahed Mikki (nahedmikki@yahoo.com)
Hanan F Abdul-Rahim (hanan.arahim@qu.edu.qa)
Faisal Awartani (faisal@alpha.ps)
Gerd Holmboe-Ottesen (gerd.holmboe-ottesen@medisin.uio.no)

Version: 11 Date: 5 December 2009

Author's response to reviews: see over
Dear editor
Thanks for accepting our manuscript for publication in BMC Public Health.
We have revised the proof that was sent to us as follows:
-We have formatted tables 3, 4 and 5 so that all the information in the tables are clear.

Regards
The authors.